Autonomous Multiagent Systems

Peter Stone Week 5, Tuesday (9/24)

Any Questions?

Pursuit Activity

Group 1: homogeneous, non-communicating Group 2: homogeneous, communicating Group 3: heterogeneous, non-communicating Group 4: heterogeneous, communicating

Overview of the Readings

• Textbook chapter 8 • Desiderata for ACLs (optional) • ACLs – The Current Landscape

Student Discussion

• Does XML replace any need for KQML/FIPA ACL?

Next Week’s Readings – Teamwork

• Joint intentions – formal model • STEAM – general teamwork construction (optional)